Author's response to reviews

Title: Do oral corticosteroids provide clinical and cost-effective symptom relief for sore throat? A multi-centre, double blind, randomized, placebo-controlled trial.

Authors:

Johanna Maughan (johanna.maughan@phc.ox.ac.uk)
Gail Hayward (gail.hayward@phc.ox.ac.uk)
Matthew Thompson (matthew.thompson@phc.ox.ac.uk)
Alastair D Hay (alastair.hay@bristol.ac.uk)
Michael Moore (mvm198@soton.ac.uk)
Paul Little (p.little@soton.ac.uk)
Kim Harman (kim.harman@bristol.ac.uk)
Jane Wolstenholme (jane.wolstenholme@dph.ox.ac.uk)
Rafael Perera (refarel.perera@phc.ox.ac.uk)
Merryn Voysey (merryn.voysey@phc.ox.ac.uk)
Julie Allen (julie.allen@phc.ox.ac.uk)
Maria Breen (maria.breen@phc.ox.ac.uk)
Carl Heneghan (carl.heneghan@phc.ox.ac.uk)

Version: 2 Date: 7 January 2014

Author's response to reviews:

Please could you modify the manuscript, ensuring that you state that you have ethical approval and give the appropriate details, and then submit a revised version by logging on to the website, http://www.biomedcentral.com/my/manuscripts.

I have added the ethical approval statement to the paper.